Author's response to reviews

Title: Task sharing of a psychological intervention for maternal depression in Khayelitsha, South Africa: Study protocol for a randomized controlled trial

Authors:

Crick Lund (crick.lund@uct.ac.za)
Marguerite Schneider (marguerite.schneider@uct.ac.za)
Thandi Davies (thandi.davies@uct.ac.za)
Memory Nyatsanza (memory.nyatsanza@uct.ac.za)
Simone Honikman (simone.honikman@uct.ac.za)
Arvin Bhana (arvin.bhana@gmail.com)
Judith Bass (jbass@jhu.edu)
Paul Bolton (pbolton@jhu.edu)
Michael Dewey (michael.dewey@kcl.ac.uk)
John Joska (john.joska@uct.ac.za)
Ashraf Kagee (skagee@sun.ac.za)
Landon Myer (landon.myer@uct.ac.za)
Inge Petersen (peterseni@ukzn.ac.za)
Martin Prince (mprinceiop@gmail.com)
Dan J Stein (dan.stein@uct.ac.za)
Graham Thornicroft (graham.thornicroft@kcl.ac.uk)
Mark Tomlinson (markt@sun.ac.za)
Atalay Alem (atalay.alem@gmail.com)
Ezra Susser (ess8@columbia.edu)

Version: 2
Date: 31 October 2014

Author's response to reviews:

We have revised the manuscript as follows:

Request 1: Please include the email addresses of all authors on the title page.
Done

Request 2: Please include the dates of registration with your trial registration numbers at the end of the Abstract.
Done.

Request 3: Please include a statement in your Methods section explaining that you obtained informed consent from each participant.
Done

Request 4: Please include a figure title and legend section after the reference list.
Done